Author's response to reviews

Title: Antiinflammatory evaluation of leaves of Plumeria acuminata

Authors:

M Gupta (malayagupta@yahoo.co.in)
Uk Mazumder (mazumderu@yahoo.co.in)
P Gomathi (pgoms@yahoo.com)
V Thamil Selvan (pharmtamil@yahoo.co.in)

Version: 4 Date: 8 September 2006

Author's response to reviews: see over
Date: 08.09.06
From,
PROF. Mrs. MALAYA GUPTA,
Division of Pharmacology,
Department of Pharmaceutical Technology,
Jadavpur University,
Kolkata.
India - 700 032.

To,
Editor-in-Chief,
BMC Complementary and Alternative medicine.

Sub: Submission of revised manuscript entitled "Antiinflammatory evaluation of leaves of Plumeria acuminata".

Dear Sir,

Here by I am sending the revised manuscript entitled "Antiinflammatory evaluation of leaves of Plumeria acuminata" along with replies to the reviewer’s comments for publication in your esteemed Journal. I would be highly obliged if you acknowledge the receipt.

Thanking you,

Yours Faithfully

Malaya Gupta
Replies to the reviewer’s comments

Reviewer No 1:

1. As per the suggestion of the learned referee the reference for statistical analysis have been included in the revised manuscript.

Reviewer No 2:

No suggestions were given by the reviewer

Reviewer No 3:

No suggestions were given by the reviewer